Author's response to reviews

Title: Efficacy and safety of a subacromial continuous ropivacaine infusion for post-operative pain management following arthroscopic rotator cuff surgery: a protocol for a randomised double-blind placebo- controlled trial.

Authors:

Jennifer A Coghlan (coghlan@bigpond.net.au)
Andrew Forbes (andrew.forbes@med.monash.edu.au)
Simon N Bell (snbell@bigpond.net.au)
Rachelle Buchbinder (rachelle.buchbinder@med.monash.edu.au)

Version: 5 Date: 20 April 2008

Author's response to reviews:

The department name is changed.
*Monash* Department of Clinical Epidemiology, Cabrini Hospital, Monash University, Melbourne

(The first Monash was accidentally omitted)